Author’s response to reviews

Title: Differences in anti-inflammatory effect of immature and mature of Rubus coreanus fruits on LPS-induced Raw 264.7 macrophages via NF-κB signal pathways

Authors:

Kyung Hye Seo (seokh@korea.kr)
Ji Lee (ljy341@chungbuk.ac.kr)
Jeong-Yong Park (yong0433@korea.kr)
Gwi Jang (janggy@korea.kr)
Hyung Don Kim (radioleg@korea.kr)
Young Lee (youngseoblee@korea.kr)
Dong Hwi Kim (kimodh@korea.kr)

Version: 4 Date: 27 Mar 2019

Author’s response to reviews:

Dear Editor

I wish to submit a research article, titled “Differences in anti-inflammatory effect of immature and mature of Rubus coreanus fruits on LPS-induced Raw 264.7 macrophages via NF-κB signal pathways.” for consideration for publication in BMC Complementary and Alternative Medicine.

1. Please include a title page as the first page of your manuscript where you list the title of the manuscript and the full names, institutional addresses and email addresses for all authors. The corresponding author should be indicated.

->We added.

2. Please note that Research article require the following sections:
- Background
- Results
- Discussion
- Conclusions
- Methods

Please ensure that these sections are present and clearly labelled as described above. Please do check the instructions for authors on the journal website to ensure that your manuscript follows the correct structure for this journal and article type, and to ensure that you are aware of additional recommendations for formatting that will facilitate handling of your manuscript.

->We edited.

3. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

Please see here for details on the information to be included in these sections:

https://bmccomplementalternmed.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article
If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations.

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

4. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

5. We would like to request you to please upload your figures as separate files.

-> We uploaded JPG files.